Author’s response to reviews

Title: Influence of Gender, Sexual Orientation, and Need on Treatment Utilization for Substance Use and Mental Disorders: Findings from the California Quality of Life Survey

Authors:

  Christine E. Grella (grella@ucla.edu)
  Lisa Greenwell (lgreenwel@ucla.edu)
  Vickie M. Mays (mays@ucla.edu)
  Susan D. Cochran (cochran@ucla.edu)

Version: 5 Date: 22 June 2009

Author’s response to reviews:

we are submitting final revised version of this manuscript